Author's response to reviews

Title: How effective is good domestic kitchen hygiene at reducing diarrhoeal disease in developed countries?: a systematic Review and reanalysis of the UK IID study

Authors:

Anna Stenberg (annastenberg80@hotmail.com)
Clare Macdonald (clare.macdonald@doctors.org.uk)
Paul R Hunter (Paul.Hunter@uea.ac.uk)

Version: 5 Date: 13 February 2008

Author's response to reviews:

Please find attached our final version. We have made all the fromatting changes as requested and especially reordered the references.

As regards the referee's comments:

1. We have not made the change as requested as the balance of evidence is against the hypothesis and so does refute it, though with our later caviats.
2. change made

In addition